Author's response to reviews

Title: Increase of Gap Junction Activities in SW480 Human Colorectal Cancer Cells

Authors:

Kristina M Bigelow (kr16b@vet.k-state.edu)
Thu A Nguyen (tnguyen@vet.k-state.edu)

Version: 2
Date: 28 March 2014

Author's response to reviews: see over
March 28, 2014

BMC Cancer
BioMed Central
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom

Re: Manuscript Revision, entitled “Increase of Gap Junction Activities in SW480 Human Colorectal Cancer Cells”

Dear Editorial Board,

The manuscript has been revised to include line and page numbers. An ethics statement has been added in the methods section. The figures of the manuscript have been revised to meet with the 300 dpi standards.

Sincerely,

Kristina M. Bigelow

Kristina M. Bigelow